

### Large bottleneck size in Cauliflower Mosaic Virus populations during host plant colonization

Baptiste Monsion, Rémy Froissart, Y. Michalakis, Stéphane Blanc

### ▶ To cite this version:

Baptiste Monsion, Rémy Froissart, Y. Michalakis, Stéphane Blanc. Large bottleneck size in Cauliflower Mosaic Virus populations during host plant colonization. PLoS Pathogens, 2008, 4 (10), 10.1371/journal.ppat.1000174 . hal-02668679

### HAL Id: hal-02668679 https://hal.inrae.fr/hal-02668679v1

Submitted on 31 May 2020  $\,$ 

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

### Large Bottleneck Size in *Cauliflower Mosaic Virus* Populations during Host Plant Colonization

### Baptiste Monsion<sup>1</sup>, Rémy Froissart<sup>1,2</sup>, Yannis Michalakis<sup>2</sup>, Stéphane Blanc<sup>1</sup>\*

1 UMR BGPI, INRA-CIRAD-SupAgroM, Campus International de Baillarguet, Montpellier, France, 2 GEMI, UMR CNRS-IRD 2724, IRD, Agropolis, Montpellier, France

### Abstract

The effective size of populations (*Ne*) determines whether selection or genetic drift is the predominant force shaping their genetic structure and evolution. Despite their high mutation rate and rapid evolution, this parameter is poorly documented experimentally in viruses, particularly plant viruses. All available studies, however, have demonstrated the existence of huge within-host demographic fluctuations, drastically reducing *Ne* upon systemic invasion of different organs and tissues. Notably, extreme bottlenecks have been detected at the stage of systemic leaf colonization in all plant viral species investigated so far, sustaining the general idea that some unknown obstacle(s) imposes a barrier on the development of all plant viruses. This idea has important implications, as it appoints genetic drift as a constant major force in plant virus evolution. By co-inoculating several genetic variants of *Cauliflower mosaic virus* into a large number of replicate host plants, and by monitoring their relative frequency within the viral population over the course of the host systemic infection, only minute stochastic variations were detected. This allowed the estimation of the CaMV *Ne* during colonization of successive leaves at several hundreds of viral genomes, a value about 100-fold higher than that reported for any other plant virus investigated so far, and indicated the very limited role played by genetic drift during plant systemic infection by this virus. These results suggest that the barriers that generate bottlenecks in some plant virus species might well not exist, or can be surmounted by other viruses, implying that severe bottlenecks during host colonization do not necessarily apply to all plant-infecting viruses.

Citation: Monsion B, Froissart R, Michalakis Y, Blanc S (2008) Large Bottleneck Size in Cauliflower Mosaic Virus Populations during Host Plant Colonization. PLoS Pathog 4(10): e1000174. doi:10.1371/journal.ppat.1000174

Editor: Ralph S. Baric, University of North Carolina, United States of America

Received April 16, 2008; Accepted September 12, 2008; Published October 10, 2008

**Copyright:** © 2008 Monsion et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was funded by an INRA grant JE77, and a PhD grant for BM from Region Languedoc-Roussillon and INRA SPE. YM acknowledges support from CNRS and IRD.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: blanc@supagro.inra.fr

### Introduction

The main forces driving changes of the frequencies of alleles within populations are selection (deterministic) and genetic drift (stochastic). While both forces obviously act concomitantly in most cases, the balance of their respective action is governed by a number of factors, one of the most important being the effective population size (Ne) [1–3]. This parameter, commonly defined as the size of an ideal population which would drift at the same rate as the observed population [4], is widely investigated in the literature both theoretically [5-8] and experimentally [9-11] in a broad variety of organisms. Ne significantly informs whether the evolution of a given population might be better described by a deterministic or a stochastic evolution model [12-14]. In brief, when Ne is large, competition between genetic variants is fully acting, with no or little interference of random processes, and selection shapes the genetic composition of corresponding populations. Conversely, in populations with a small Ne, various processes resulting in stochastic "sampling" of genetic variants that will engender the next generation are prevalent and counter the effect of selection.

Experimental evolution using viruses as biological models has developed enormously during the past two decades. A series of converging reports on viruses of bacteria, animals, and plants confirmed that protocols maintaining constant large population sizes lead to selection of the fittest variants, thus augmenting the mean fitness of viral populations [15–17]. On the other hand, the repeated experimental imposition of severe bottlenecks (temporarily reducing Ne is associated with relaxed selection and strong drift, dramatically reducing the mean fitness in the population [18-21]. Viruses are intuitively perceived as having extremely large population sizes during host invasion. While this is true for the census population size, i.e. the number of viral genomes accumulating in single hosts, it does not indicate what fraction thereof is actually actively replicating and propagating, or is efficiently progressing into new organs and tissues, yielding subsequent generations upon host colonization. Estimating viral Ne during systemic infection of a host has proved experimentally difficult, as illustrated by the impressive number of studies published, and still debated, on HIV [22]. Consequently, virus species for which Ne has been formally evaluated during invasion of various organs and tissues of the host remain extremely few in both plants and animals. The question of whether deterministic or stochastic models better explain the evolution of viral populations is thus pending further investigation of Ne.

In plants, several specific barriers could impose severe bottlenecks on populations of invading viruses. The level at which the existence of such bottlenecks has been most documented is the colonization of new leaves, not only because of the obvious practical ease of comparing the genetic content of different leaves from the same host, but also because this level results from longdistance movement of the virus population, which is loaded into the vascular system from source leaves and unloaded into sink

### Author Summary

Whether selection or stochastic genetic drift is the major force driving the evolution of a virus depends largely on the size of the viral population, with the former being predominant in large populations and the latter taking over when population sizes are transiently or durably reduced. This question has been intensively debated in both plant and animal viruses, as demographic fluctuations throughout viral life cycles are poorly understood. In plant viruses, an extremely small population size-down to a few founder genome units colonizing each leaf-has been formally estimated in two instances, and all other virus species investigated so far have consistently been shown to undergo extreme demographic bottlenecks during systemic invasion of their host. This situation conveys the general idea that all viruses are confronted with "universal barriers" in plants, imposing repeated transient decreases in their population size, thus making genetic drift a major constant driver of their evolution. Here, using the example of Cauliflower mosaic virus, we mitigate this general idea by showing that at least one virus species can overcome such putative limiting barriers and massively invade leaves with hundreds to thousands of founding genome units.

leaves (for review, see [23]). As discussed later, other phenomena could also generate population bottlenecks, but virus movement within the host plant through plasmodesmata is most often considered as a major putative obstacle to exponential expansion.

While not directly estimating Ne, two experimental evolution studies demonstrated the existence of viral population bottlenecks during leaf colonization. With a mixture of 12 engineered genetic variants of *Cucumber mosaic virus* (CMV) co-inoculated into tobacco plants, Li and Roossinck [24] showed that diversity decreased randomly but continuously in the viral population, as increasing numbers of variants were lost when systemic infection progressed into upper, newly-formed leaves. Even more illustrative was the spatial distribution of the genetic diversity of a *Plum pox virus* (PPV) population, maintained in a perennial host tree for over 13 years [25]. While a large number of related PPV variants could be distinguished and detected in various organs and tissues, distinct subpopulations were shown to be isolated in different branches. Extreme population bottlenecks were further evidenced when the virus progressed into newly formed leaves, which were all colonized by one single viral genotype.

The effective size of virus populations during systemic infection of host plants has been evaluated more formally in two instances. Populations of *Wheat streak mosaic virus* (WSMV) invading a wheat tiller [26], and of *Tobacco mosaic virus* (TMV) invading a tobacco leaf [27], were shown to be founded by as few as 4 and 2–20 virus particles, respectively, hence again resulting in severe bottlenecks.

All the studies cited above converged to the conclusion that the population size of plant viruses fluctuates dramatically and can be temporarily remarkably small, i.e. of the order of one to a few genome units founding the population that subsequently develops to billions of genomes in each systemically infected leaf. Such a demographic regime suggests that genetic drift is a major force in plant virus evolution, as proposed and discussed by several authors [24–28]. The fact that extreme population bottlenecks are consistently described for unrelated virus species infecting mono-or di-cotyledonous hosts appealingly suggests that all plant viruses might be subject to the same phenomenon [24,27–29], perhaps related to the unavoidable physical barriers that hamper the

systemic movement of all viruses in plants. However, this tempting generalisation definitely requires closer inspection as, if proven true, it would illustrate the impossibility of viral adjustments and trade-offs on important traits requiring large Ne. In contrast, a single counter example would demonstrate that, in some virusplant associations, such barriers might not exist or might be surmounted by the virus. Further research on more diverse virus species, with different replication strategies and different life cycles, is still needed in order to assume extremely small Ne as a general rule during host plant colonization.

In this report, we assessed the importance of bottlenecks during systemic host colonization in populations of Cauliflower mosaic virus (CaMV), a DNA virus whose biological properties differ largely from those of the RNA viruses investigated in the studies cited above. Monitoring the frequency of several engineered allelic variants within leaves of single host plants revealed remarkably small stochastic fluctuations in the genetic structure of numerous CaMV populations, even over considerable periods of time. Beyond demonstrating that CaMV populations are not subject to intense genetic drift, and hence do not undergo severe demographic bottlenecks, the slight stochastic fluctuations detected were exploited to infer the effective size of CaMV populations upon systemic leaf colonization. Our estimates consistently indicate that several hundreds of genome units founded the viral populations in all leaves analyzed (originating from different plants). This value, which is ~100-fold higher than values previously reported for other plant viruses, demonstrates that extremely small viral Ne during host plant colonization is not a general rule and opens up the possibility of trade-offs on viral traits that directly or indirectly depend on Ne.

### **Materials and Methods**

#### Engineered CaMV variants

The six plasmids (pCa-VIT1 to pCa-VIT6) used in this study to generate the six CaMV allelic variants have been described in detail and characterized previously [30].

All are infectious full-length clones of the CaMV Cabb-S isolate [31], where a specific genetic marker (a dsDNA non-coding sequence of 40 bp) has been inserted between CaMV ORFs II and III. These markers should not affect any viral function, as non-coding sequences between CaMV ORF do not affect translation. When inoculated individually into turnip plants, each CaMV clone (CaMV-VIT1 to CaMV-VIT6) induces symptoms similar to CaMV wild type, and all six genetic markers have been shown to be stably maintained in the viral genome even after three successive passages in plants [30].

# Plant growth conditions and inoculation with mixed CaMV-VIT populations

All turnip plants (*Brassica rapa* var. "Just Right") were maintained in an insect-proof greenhouse under controlled conditions: 25/19°C day/night with a photoperiod of 16/8 h day/night.

Plants infected for 21 days with one of the CaMV-VIT1 to -VIT6 clones (one plant per clone) were used to prepare virus particle-enriched fractions as previously described [30]. Equal volumes of each of the six virus particle-enriched fractions were pooled to produce a mixed inoculum (designated Mix6VIT) containing all six allelic variants of CaMV. Fifty healthy young plantlets, at the three-leaf stage, were mechanically infected by rubbing 20  $\mu$ l of Mix6VIT solution onto the entire surface of the two larger leaves, previously powdered with abrasive carborundum. First symptoms indicative of CaMV infection appeared on non-inoculated leaves within 7 to 9 days, and all plants proved systemically infected at 11 days post-infection (dpi).

#### Harvesting of leaf samples

At 13 dpi, i.e. very soon after systemic infection had developed, the inoculated leaves were discarded and all large expanded leaves were collected. In all cases, two to three young expanding or emerging leaves in the centre of the rosette were left unsevered in order to allow continuous growth of the plants for further sampling as described below. The viral population extracted from each plant at this early stage was considered as the initial population.

Thirty-two days later (at 45 dpi) the same plants had grown continuously and produced 10 to 15 newly expanded leaves. At this stage, one single leaf from each plant, randomly chosen between the 5<sup>th</sup> and the 12<sup>th</sup> newly formed leaves, was collected. The viral population extracted from each single leaf at this late stage was considered as the final population. All leaf samples were stored at  $-20^{\circ}$ C until further viral DNA purification.

The rationale of this sampling protocol is further discussed below.

### Purification of viral DNA and genetic composition analysis

Purification of viral DNA from the harvested leaf samples, as well as the analysis of the genetic structure of the corresponding viral genome populations were performed exactly as described previously, using the QSS (Quantitative Single-letter Sequencing) method [30]. Briefly, QSS allows the simultaneous quantification of numerous allelic variants in a single DNA sample. After a PCR amplification step using a pair of primers flanking the markercontaining region, the PCR product is submitted to single-letter sequencing primed with a fluorescently-labelled oligonucleotide, located upstream of the markers' position. The resulting monochromatic electropherogram exhibits numerous specific diagnostic peaks, attributable to specific variants, signifying their presence/absence in the DNA sample. Finally, peak fluorescence can be quantified and used to estimate the frequency of the corresponding variant in the DNA population. The accuracy and reproducibility of the QSS method have been fully evaluated and shown to be equivalent or higher than that of competing technologies (including those based on real-time PCR) for quantifying variants with a relative frequency above 5% in the DNA population [30].

#### Statistical analysis

Changes in the relative frequencies of the six alleles (-VIT1 to -VIT6) between the initial and final populations, sampled as described above, were precisely monitored in 50 replicate test plants.

In order to calculate the number of founder genomes initiating each final population in a single leaf (N), all parameters accounting for the transition from p (initial frequency of a given marker) to p' (final frequency of the same marker) must be evaluated. For any given plant the difference between p and p' ( $\Delta p$ ) may be potentially attributed to genetic drift and to selection:

$$\Delta p = \Delta p \operatorname{Drift} + \Delta p \operatorname{Selection}$$
(1)

There is no a priori reason to assume that any selection affecting the frequency of the markers is heterogeneous across plants (e.g. favouring a marker in a plant and selecting against it in another plant). Moreover, given the way our markers were constructed, there is no *a priori* reason to expect directional selection in favour of any of them. For the moment, we will thus assume that selection is negligible, and the markers effectively neutral, and will provide further arguments supporting this assumption at the end of this section.

We used two methods to estimate N, both based in the change of genetic variance between the two sampling events.

The first method directly tracks changes in variance. The variance of the estimates of p and p' between plants can be written as:

$$Var (p') = Var (p) + Var (Drift)$$
(2)

The variance due to drift is equal to:

$$Var (Drift) = p(1-p)/N$$
(3)

Where p is the frequency of a given marker in initial populations. From Equations 2 and 3 we obtain:

$$N = \frac{p(1-p)}{Var(p') - Var(p)} \tag{4}$$

The second technique is based on Fst statistics. Fst statistics were introduced by Wright [4] and represent a way to partition genetic variance within and between populations. One way to express Fst is:

$$Fst = (H_T - H_S)/H_T$$
(5)

where  $H_T$  is gene diversity assuming all populations form a single large population and  $H_S$  is the average gene diversity within each population. In our case, each plant represents a population. Gene diversities express the probability to randomly draw two different alleles and are thus equal to  $1 - \sum_i p_i^2$ , where  $p_i$  is the frequency of each allele at the subdivision level under consideration. In our case  $H_T$  represents the gene diversity obtained after calculating the average frequency of each marker across all plants, while  $H_S$ represents the average across all plants of within plants gene diversities

Using standard population genetics theory (e.g. [32]) it can be shown that, for a haploid such as a virus,

$$F_{ST}^{\prime} = \frac{1}{N} + \left(1 - \frac{1}{N}\right)F_{ST} \tag{6}$$

where the prime denotes sampling at two different points in time. Rearranging Equation 6 yields an expression that can be used to estimate N:

$$N = \frac{1 - F_{ST}}{F'_{ST} - F_{ST}}$$
(7)

We will refer to the estimates obtained by Equations 4 and 7 as  $N_{\rm V}$  and  $N_{\rm F}$  respectively.

To provide a confidence interval for these estimates we used a resampling technique. We bootstrapped over plants: for each bootstrap we randomly drew a sample of 50 plants with replacement, calculated Var(p'), Var(p) and p, and from that estimated  $N_V$ . We also calculated  $H_T$  and  $H_S$ , and from that estimated  $N_F$ . We repeated this procedure 1,000 times and constructed a distribution of  $N_V$  and  $N_F$ . The 95% confidence intervals correspond to the 2.5 and 97.5 percentiles of these distributions. In several bootstraps, the final variance was smaller

than the initial variance, or the final  $F_{ST}$  was smaller than the initial  $F_{ST}$ , yielding negative values of  $N_V$  or  $N_F$ . In such cases, our estimation method does not work. Such cases would correspond to conditions leading to convergence of marker frequencies across plants, i.e. conditions where drift is negligible, and thus N can be regarded as infinite. We conservatively discarded such cases from the calculation of the upper limit of the confidence interval.

The average changes in frequency of the markers proved very small (see Table 1). Neutrality tests applied to VIT1, 3 and 4, demonstrated that only VIT1 and 3 might be slightly selected for and against, respectively (data not shown). Nevertheless, both these variants were used for estimating  $\mathcal{N}$  because, while directional selection would lead to a situation where marker frequencies would be more similar across plants at the second sampling event than at the first, yielding smaller final variance and  $F_{ST}$ , our data show the opposite, strongly contradicting the directional selection hypothesis. Another *a posteriori* justification is that while the neutrality tests indicate that one marker is slightly selected for, another is slightly selected against and another is effectively neutral, the  $N_V$  estimates obtained form all three are of the same order of magnitude. These arguments suggest that selection, if present, can be neglected relative to drift in our experiment.

### Results

### Genetic structure of initial CaMV populations in whole plants

Following systemic spread of infection in plants inoculated with a mixture of 6 CaMV genetic variants (Mix6VIT), populations of viral genomes in each plant were purified from all fully expanded and systemically infected leaves, excluding inoculated leaves. DNA samples from each of 50 replicate plants were considered as the initial populations and were submitted to QSS analysis to detect markers in CaMV genomes and quantify their relative frequency as described in Materials and Methods. The mean frequency of each marker and the variance among the 50 CaMV populations collected at this stage is given in Table 1 (full dataset with details for each plant is provided in Table S1).

**Table 1.** Genetic structure of initial and final CaMV populations.

	Initial populations <sup>a</sup>	Final populations <sup>a</sup>	E(∆p) <sup>d</sup>		
	$p^b \times 10^2$ (SD <sup>c</sup> . 10 <sup>2</sup> )	p' <sup>b</sup> ×10 <sup>2</sup> (SD <sup>c</sup> . 10 <sup>2</sup> )			
VIT1	41.6 (3.52)	45.8 (4.27)	4.2		
VIT2	4.2 (1.90)	3.3 (1.55)	-0.9		
VIT3	42.5 (4.19)	40.9 (5.08)	-1.6		
VIT4	10.2 (2.88)	9.2 (3.19)	-1.0		
VIT5	1.7 (1.89)	1.9 (1.61)	0.2		
VIT6	1.7 (0.98)	3.6 (1.70)	1.9		

The genetic structure of the 50 sampled CaMV populations is determined by the different markers present in each, as well as by their respective relative frequency. For reasons of clarity, only mean frequency values and standard deviations among the 50 repeats are shown here. The full dataset with details of the 50 CaMV populations is available (Table S1).

<sup>a</sup>Sampling protocol for initial and final populations is described in Materials and Methods.

<sup>b</sup>Mean relative frequency of each marker over the 50 replicates.

<sup>c</sup>Standard deviation of marker frequencies across the 50 replicates.

<sup>d</sup>Mean difference between initial and final frequencies in each replicate, and for each marker.

All values are expressed as percent of the viral genome population.

doi:10.1371/journal.ppat.1000174.t001

The six CaMV-VIT genotypes were found in widely unequal ratios in all 50 cases, with mean relative frequencies being close to 0 for CaMV-VIT5 and -VIT6, and up to around 40% for CaMV-VIT1 and -VIT3. However, a striking observation was that each marker was found with a very similar frequency in all 50 replicate CaMV populations (Table S1), as demonstrated by the very low standard deviation among repeats (Table 1). This result clearly indicates that little stochastic variation is generated by the inoculation process, or between the inoculation process and the sampling of initial populations. The frequency differences between markers being very consistent in all 50 replicates, they are certainly due to a non-stochastic phenomenon (discussed below).

# Genetic structure of final CaMV populations in single leaves

At late infection stages, 32 days after collecting the initial populations, a single systemically infected leaf per individual plant, randomly chosen between the 5<sup>th</sup> and the  $12^{th}$  new expanded leaves, was harvested. The CaMV populations extracted from these sampled leaves are considered as "final populations," and their genetic structure (Table 1 and S1) was evaluated in exactly the same manner as that of the initial populations.

Remarkably, in all 50 repeats, the final population resembled the corresponding ancestor population despite the different sampling process (individual leaves for final populations versus pools of leaves for initial populations), demonstrating that very little change had occurred over this considerable time period, whatever the position or age of the analyzed leaf (see  $E(\Delta p)$  values [Table 1], and compare p and p' values [Table S1] for each of the 50 plants). Consequently, as between initial populations, there were only small variations between the final populations, although the standard deviation was slightly higher in the latter case [Table 1].

Taken together, these results argue for the absence of large stochastic variations in the genetic structure of a CaMV population upon progression of the infection into newly formed leaves. This suggests that the effective size (Ne) of CaMV populations during host systemic colonization is likely to be surprisingly large (compared to data previously published on other plant viruses) as specifically evaluated in the next section.

## Estimation of bottleneck size in CaMV populations during systemic leaf colonization

Because the average relative frequency of markers CaMV-VIT2, -VIT5 and -VIT6 was close to, or even below, the limit of accuracy of the QSS method [30], their quantification was considered poorly reliable, and hence these markers were not used for further analysis. We thus used only the three markers -VIT1, -VIT3 and -VIT4 to estimate the average number of genomes founding the population in each leaf (N), as described in the Materials and Methods.

The estimates of N from these three markers yielded remarkably high values, corresponding to several hundreds of viral genomes. Two different statistical methods were applied to the data set and provided very consistent results, both for the observed bottleneck size and for the limit of the 95% confidence interval (compare  $N_V$  in Table 2, and  $N_F$  in Table 3). The lower limit of the confidence interval was >100 in all cases, whereas the higher limit reached thousands.

### Discussion

This report evaluates the effective size of CaMV populations during systemic invasion of plant leaves. Several previous studies have used distinct experimental protocols to tackle similar **Table 2.** Number of CaMV genomes founding the population in individual leaves, estimated through the analysis of the variance of the frequency of markers.

Marker <sup>a</sup>	Var(p')×10 <sup>4</sup>	Var(p)×10 <sup>4</sup>	Nv <sup>b</sup>	95% confidence interval
VIT1	18.13	12.39	423.00	180–5034
VIT3	25.79	17.60	298.45	122-3219
VIT4	10.19	8.30	484.61	166–5193

 $^{\rm a}\textsc{Only}$  markers VIT1, VIT3, and VIT4 were used for estimating Nv, for reasons explained in the text.

 ${}^{b}N_{v}$  (and 95% confidence interval) is calculated according to Equation 4 (see Materials and Methods), where p is estimated from values given in Table 1. doi:10.1371/journal.ppat.1000174.t002

**Table 3.** Number of CaMV genomes founding the population

 in individual leaves, estimated through Fst statistics.

	Initial populations <sup>a</sup>	Final populations <sup>a</sup>
H <sub>T</sub>	0.63530089	0.61474821
Hs	0.6314719	0.6093363
Fst	0.006027	0.008803
N <sub>F</sub> =358 (184–1909) <sup>b</sup>		

 $^{\rm a}\text{Only}$  markers VIT1, VIT3, and VIT4 were used for estimating  $N_{\text{F}}$ , for reasons explained in the text.

 ${}^{p}N_{F}$  was estimated from Fst values using Equation 7 (see Materials and

Methods). The confidence interval, given in parentheses, was estimated by bootstrapping over plants.

doi:10.1371/journal.ppat.1000174.t003

questions with various viral species; the specifics and rationale of the protocol used here are discussed below. To eliminate variations in the genetic structure of viral populations that could be related to the inoculation process, the virus population present in the whole plant (excluding the inoculated leaves) soon after systemic spread of infection was considered as the starting point of the experiment, putative subsequent changes thus occurring only under the influence of within-plant processes. At this initial stage, the virus population uploaded into, and circulating within, the vascular system is most likely best represented by the overall content of the systemically infected leaves, as they have either received viruses from the vasculature, exported viruses into it, or both. Expanded infected leaves were therefore harvested for initial analysis, carefully preserving two to three young newly expanding leaves on the still-growing plant. The infected plants were then left to grow for a period of 32 days, during which the virus population successively colonized 10 to 15 emerging and expanding new leaves. Regardless of where the virus population originates from during this process (vascular system and roots at the beginning, then increasing numbers of leaves later on), analysis of single leaf (between positions 5 and 12 above the initial harvest) contents collected 32 days later should reveal the existence of any putative bottlenecks at any stage of the systemic infection. Indeed, final CaMV populations result largely from the successive leaf-to-leaf passages that occur sequentially when young sink leaves are infected, become sources and subsequently export virus into new sinks.

All genetic variants (CaMV-VIT1 to CaMV-VIT6) were coinoculated at similar locations and at the same time point onto the two first true leaves of turnip plantlets. The reason why the relative proportions of the six variants are highly unequal later in systemically infected plants was mostly that unequal proportions were already present in the initial inoculum, Mix6VIT (further discussed in Table S2). Previous studies, based on co-infection of turnip plants by two distinct CaMV variants with seemingly equal growth rate, have compared the variant ratios in the initial inoculum and in resulting systemically infected plants. No differences were observed when concentrated virus particles were used for inoculation [33], whereas stochastic fluctuations were detectable when inoculum consisted in viral DNA prepared from infectious clones [34], presumably due to the lower infectivity of DNA preparations, engendering a stochastic founding effect in the latter case. These two studies indicate that indeed the inoculation process could induce unwanted fluctuations in repeated inoculations and prompted us to use virus particles enriched preparation for the purpose of our experiment (albeit development of systemic infection in between the two observation time points chosen in our protocol should not be affected by inoculation variations). Consistently, all 50 test plants in our study contained similar ratios of the six variants, suggesting very little stochastic variation during the inoculation process.

The values of p' were estimated from individual leaf samples, each collected on a different test plant at a random position (between leaf position 5 and 12 above the initial harvest point). The remarkable observation that only relatively small variations in p' are recorded in leaves from different plants (see standard deviation in Table 1) strongly indicates that very little variation occurs between leaves of the same plant. We therefore conclude that a similarly large founding population of several hundred viral genomes colonizes each newly formed leaf in plants systemically infected by CaMV.

The number of founder CaMV genomes in systemically infected leaves appears to be at least 100-fold higher than that determined in previously published studies for various RNA viruses [25-27]. Although there are clear differences in the inoculation and sampling protocols described in different studies, the presence of severe bottlenecks in WSMV [26], TMV [27], CMV [24] and PPV [25], and their obvious absence in CaMV, can hardly be explained by artifacts of experimental designs. Indeed, at least in the present study and in three of the cited examples [24–26], viruses moved not only from inoculated to systemic tissues, but also from systemically infected tissues to newly formed leaves or tiller. As mentioned in the Introduction, the previous repeated demonstration that different viruses undergo extreme bottlenecks during systemic infection of their host plant suggested that unavoidable barriers (e.g. connections between cells, etc.) may exist similarly for all plant viruses. The results presented here clearly demonstrate that such putative limiting barriers can be surmounted by some viruses and that the size of the viral population circulating in planta might thus be directly or indirectly controlled and needs to be evaluated for each virus individually. Whether the absence of severe bottlenecks demonstrated here for CaMV is common, and whether this phenomenon is related to the biology of the host plant, that of the virus, or more intricately on specific virus-host associations remains an open question.

The situation described here for CaMV re-opens the question as to what phenomena actually generate bottlenecks during virus infection. A reasonable hypothesis explaining the presence or absence of severe demographic bottlenecks would be the regulation of multiple infection of cells by several genomes of the viral population, in other words, regulation of the multiplicity of infection (MOI) of host cells. Indeed, it was proposed recently that bottlenecks seen in a PPV population infecting a *Prunus* tree could stem from the fact that viral genomes cannot secondarily invade tissues that have already been infected by a closely related genome [25], thus preventing extensive mixing of genetic variants within host. Consistently, two isogenic PPV genomes, but expressing fluorescent proteins of different colours, revealed their mutual exclusion within infected tissues, with apparently rare coinfection of single cells [35], a phenomenon also reported for other virus species, such as WSMW [26], TMV [36], and a few other RNA viruses [37]. This phenomenon could logically induce a very low MOI, in turn engendering bottlenecks (and small Ne), due to competition for host "territories" between variants of the same population. Accordingly, under this hypothesis, the absence of severe bottlenecks found for CaMV would suggest a higher MOI that is totally consistent with the remarkably high within-host recombination rate described for this virus [38], recombination being possible solely in multiply infected cells. We believe that explaining, at least in part, variations of within-plant bottlenecks in different virus species by their different capacity to multiply infect host cells is a very appealing hypothesis that will stimulate further novel research, the "natural" MOI during infection of a multi-cellular host being virtually unknown in viruses of both animals and plants.

### **Supporting Information**

**Table S1** Whole dataset from the 50 replicate infected plants. p represents the marker relative frequency in initial populations. p' represents the marker relative frequency in final populations. All values are indicated as percentage of the viral genome population. -, marker not detected.

Found at: doi:10.1371/journal.ppat.1000174.s001 (127 KB DOC)

**Table S2** Analysis of the initial inoculum Mix6VIT. <sup>a</sup>The mixture Mix6VIT was prepared and analysed by QSS as described in Materials and Methods. <sup>b</sup>p<sub>0</sub> = mean relative frequency determined from five independent repeats of QSS measurement. <sup>c</sup>Standard deviation calculated from the five repeated QSS analysis. All values are expressed as percent of the viral genome population. The distribution of the relative frequencies of all

### References

- Nei M, Maruyama T, Chakaraborty R (1975) The bottleneck effect and genetic variability in populations. Evolution 29: 1–10.
- Gherman A, Chen PE, Teslovich TM, Stankiewicz P, Withers M, et al. (2007) Population bottlenecks as a potential major shaping force of human genome architecture. PLoS Genet 3: e119. doi:10.1371/journal.pgen.0030119.
- Kimura M, Maruyama T, Crow JF (1963) The mutation load in small populations. Genetics 48: 1,303–1,312.
- 4. Wright S (1931) Evolution in Mendelian populations. Genetics 16: 97-159.
- Criscione CD, Blouin MS (2005) Effective sizes of macroparasite populations: a conceptual model. Trends Parasitol 21: 212–217.
- Nei M, Tajima F (1981) Genetic drift and estimation of effective population size. Genetics 98: 625–640.
- Tajima F, Nei M (1984) Note on genetic drift and estimation of effective population size. Genetics 106: 569–574.
- Crow JF, Kimura M (1972) The effective number of a population with overlapping generations: a correction and further discussion. Am J Hum Genet 24: 1–10.
- Araki H, Waples RS, Blouin MS (2007) A potential bias in the temporal method for estimating Ne in admixed populations under natural selection. Mol Ecol 16: 2,261–2,271.
- Wang J (2005) Estimation of effective population sizes from data on genetic markers. Philos Trans R Soc Lond B Biol Sci 360: 1,395–1,409.
- Johnson JA, Bellinger MR, Toepfer JE, Dunn P (2004) Temporal changes in allele frequencies and low effective population size in greater prairie-chickens. Mol Ecol 13: 2,617–2,630.
- Rouzine IM, Rodrigo A, Coffin JM (2001) Transition between stochastic evolution and deterministic evolution in the presence of selection: general theory and application to virology. Microbiol Mol Biol Rev 65: 151–185.
- Jain K, Krug J (2007) Deterministic and stochastic regimes of asexual evolution on rugged fitness landscapes. Genetics 175: 1,275–1,288.
- Wahl LM, Krakauer DC (2000) Models of experimental evolution: the role of genetic chance and selective necessity. Genetics 156: 1,437–1,448.
- Weinreich DM, Chao L (2005) Rapid evolutionary escape by large populations from local fitness peaks is likely in nature. Evolution Int J Org Evolution 59: 1,175–1,182.

VIT1-6 variants in the initial Mix6VIT inoculum was globally similar to that in the 50 inoculated plants (VIT1~VIT3> VIT4>VIT2>VIT5≈VIT6). Nevertheless, apart from VIT1 and VIT3 which increased by about 10% in frequency from inoculum to infected plants, all other variants equally decreased by approximately 5% (compare  $p_0$  values in Table S2 and p values in Table 1). This phenomenon remains unclear and could be due to several different explanations, as for example: i) selection acting specifically at inoculation, could favour or disfavour some of the variants, ii) an undetermined threshold effect at the inoculation step could have positively and negatively affected the most and less frequent variants, respectively; iii) each variant in Mix6VIT originating from a different plant extract, they might have been differentially infectious due to unwanted and unequal damages of virus particles during extraction. Nevertheless, it is important to note that these considerations concern only the inoculation step, as only very minute changes in the mean frequency of all markers were detected later, over the 32 days separating the initial and final populations (see  $E(\Delta p)$  values in Table 1).

Found at: doi:10.1371/journal.ppat.1000174.s002 (32 KB DOC)

#### Acknowledgments

We are grateful to one anonymous referee for suggesting the Fst estimation method. We thank three referees for helpful comments and constructive criticism. Turnip seeds were kindly provided by the Takii Seed Company (Japan). We are thankful to the CaGeTE group for critical reading and discussion of the manuscript and to Jean-Luc Macia for all greenhouse work, plant production, and maintenance. YM acknowledges support from CNRS and IRD.

#### **Author Contributions**

Conceived and designed the experiments: BM RF YM SB. Performed the experiments: BM RF. Analyzed the data: BM RF YM SB. Wrote the paper: BM YM SB.

- Willi Y, Fischer M (2005) Genetic rescue in interconnected populations of small and large size of the self-incompatible Ranunculus reptans. Heredity 95: 437–443.
- Clarke DK, Duarte EA, Moya A, Elena SF, Domingo E, et al. (1993) Genetic bottlenecks and population passages cause profound fitness differences in RNA viruses. J Virol 67: 222–228.
- Chao L (1990) Fitness of RNA virus decreased by Muller's ratchet. Nature 348: 454–455.
- de la Iglesia F, Elena SF (2007) Fitness declines in tobacco etch virus upon serial bottleneck transfers. J Virol.
- Novella IS, Quer J, Domingo E, Holland JJ (1999) Exponential fitness gains of RNA virus populations are limited by bottleneck effects. J Virol 73: 1,668–1,671.
- Yuste E, Sanchez-Palomino S, Casado C, Domingo E, Lopez-Galindez C (1999) Drastic fitness loss in human immunodeficiency virus type 1 upon serial bottleneck events. J Virol 73: 2,745–2,751.
- Kouyos RD, Althaus CL, Bonhoeffer S (2006) Stochastic or deterministic: what is the effective population size of HIV-1? Trends Microbiol 14: 507–511.
- Waigmann E, Heinlein M (2007) Viral transport in plants. Robinson DG, ed. Berlin/Heidelberg: Springer-Verlag. 188 p.
- Li H, Roossinck MJ (2004) Genetic bottlenecks reduce population variation in an experimental RNA virus population. J Virol 78: 10,582–10,587.
- Jridi C, Martin JF, Marie-Jeanne V, Labonne G, Blanc S (2006) Distinct viral populations differentiate and evolve independently in a single perennial host plant. J Virol 80: 2,349–2,357.
- French R, Stenger DC (2003) Evolution of Wheat streak mosaic virus: dynamics of population growth within plants may explain limited variation. Annu Rev Phytopathol 41: 199–214.
- Sacristan S, Malpica JM, Fraile A, Garcia-Arenal F (2003) Estimation of population bottlenecks during movement of tobacco mosaic virus in tobacco plants. J Virol 77: 9,906–9,911.
- Garcia-Arenal F, Fraile A, Malpica JM (2003) Variation and evolution of plant virus populations. Int Microbiol 6: 225–232.
- French R, Stenger DC (2005) Population structure within lineages of Wheat streak mosaic virus derived from a common founding event exhibits stochastic variation inconsistent with the deterministic quasi-species model. Virology 343: 179–189.

- Monsion B, Duborjal H, Blanc S (2008) Quantitative Single-letter Sequencing: a method for simultaneously monitoring numerous known allelic variants in single DNA samples. BMC Genomics 9: 85.
- Franck A, Guilley H, Jonard J, Richards K, Hirth L (1980) Nucleotide sequence of cauliflower mosaic virus DNA. Cell 21: 285–294.
- Whitlock MC (1992) Temporal fluctuations in demographic parameters and the genetic variance among populations. Evolution 46: 608–615.
- Zhang XS, Melcher U (1989) Competition between isolates and variants of cauliflower mosaic virus in infected turnip plants. J Gen Virol 70: 3,427–3,437.
- Riederer MA, Grimsley NH, Hohn B, Jiricny J (1992) The mode of cauliflower mosaic virus propagation in the plant allows rapid amplification of viable mutant strains. J Gen Virol 73(Part 6): 1449–1456.
- Dietrich C, Maiss E (2003) Fluorescent labelling reveals spatial separation of potyvirus populations in mixed infected Nicotiana benthamiana plants. J Gen Virol 84: 2,871–2,876.
- Giritch A, Marillonnet S, Engler C, van Eldik G, Botterman J, et al. (2006) Rapid high-yield expression of full-size IgG antibodies in plants coinfected with noncompeting viral vectors. Proc Natl Acad Sci U S A 103: 14,701–14,706.
- Takahashi T, Sugawara T, Yamatsuta T, Isogai M, Natsuaki T, et al. (2007) Analysis of the spatial distribution of identical and two distinct virus populations differently labeled with cyan and yellow fluorescent proteins in coinfected plants. Phypathology 97: 1,200–1,206.
- Froissart R, Roze D, Uzest M, Galibert L, Blanc S, et al. (2005) Recombination every day: abundant recombination in a virus during a single multi-cellular host infection. PLoS Biol 3: e89. doi:10.1371/journal.pbio.0030089.